Author’s response to reviews

Title: The association of tumor necrosis factor superfamily 13 with recurrence of immunoglobulin A nephropathy in living related kidney transplantation

Authors:

Hyung Ah Jo (hyunga2@hanmail.net)
Seung Seok Han (hansway80@gmail.com)
Sunhwa Lee (bucibuci@hanmail.net)
Joo Young Kim (cinekid6@naver.com)
Seung Hee Yang (ysh5794@gmail.com)
Hajeong Lee (dewyhj@hanmail.net)
Jae Seok Yang (jcyjs@dreamwiz.com)
Jung Pyo Lee (nephrolee@gmail.com)
Kwon Wook Joo (junephro@gmail.com)
Chun Soo Lim (cslimjy@snu.ac.kr)
Yon Su Kim (yonsukim@snu.ac.kr)
Curie Ahn (curie@snu.ac.kr)
Jin Suk Han (jshan@snu.ac.kr)
Dong Ki Kim (dkkim73@gmail.com)

Version: 7 Date: 15 Jan 2019

Author’s response to reviews:

Dear Editor

Thank you for giving us the opportunity to revise our manuscript entitled “The association of tumor necrosis factor superfamily 13 with recurrence of immunoglobulin A nephropathy in
living related kidney transplantation.” (BNEP-D-18-00039R7) submitted for consideration for publication as a research article in the BMC nephrology.

We have checked essential editor’s comments, and we upload the figure files. We hope that it will now be acceptable for publication in the BMC nephrology.

Sincerely

January-16, 2019

Dong Ki Kim, MD, PhD